Author's response to reviews

Title: Factitious Lymphoedema - A psychiatric condition mimicking Reflex Sympathetic Dystrophy: A case report

Authors:

Nnamdi Nwaejike (justnnamdi@aol.com)
H A.P Archbold (poolerarchbold@aol.com)
Darrin Wilson (wilsondarrin@aol.com)

Version: 4 Date: 5 February 2008

Author's response to reviews: see over
Dear editor,

We thank you for your email.

We have addressed the reviewers concerns as follows:

(1) For the sake of nomenclature the case should be presented as a "distal radius fracture" as it is not a typical Colles fracture and for the core of the report it makes no difference.  

We accept that our description of the patients fracture might be misleading as it is not a typical Colles fracture on x-ray (although the mechanism of injury was typical). 

We accept the reviewer’s recommendation and have changed ‘colles fracture’ to read ‘distal radius fracture’.

(2) In the discussion the most efficient treatment of RSD should not be omitted: 

Specially since available as a nasal spray the Calcitonin treatment has become much easier.

We have included a reference for the use of Calcitonin in the treatment of RSD.

We hope these revisions are satisfactory and welcome any suggestions for improvement of the manuscript

Sincerely 

Nnamdi Nwaejike